 Vc/6-Nov-04
Week45 starting 1Nov 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 1 Nov 04  ~15.00 hrs 

	LHC SM18 Magnet Test Facility 


	7
	A1
	MBBL2098
	Prep 5 Pumping + Cool Down
	299.1

	4
	A2
	MBBL3219
	PT1 IAP@1.9K, Part III
	1.9

	3
	B1
	SSS067
	Stretched Wire Cold SC25
	1.9

	8
	B2
	_
	_
	298

	12
	C1
	MBBL2115
	ICS 5 Traveller Last
	298

	6
	C2
	MBBL1163
	Prep 5 Pumping + Cool Down
	281.8

	9
	D1
	SSS027
	ICS 3 Traveller after warm test
	296.8

	2
	D2
	SSS065
	Stretched Wire Cold SC25
	1.91

	10
	E1
	MBAL1094
	ICS 4 Final connection
	297.4

	5
	E2
	MBBL3099
	Prep 5 Pumping + Cool Down
	82.4

	11
	F1
	_
	_
	297.8

	1
	F2
	MBBR3188
	PT 12 Warm Up
	204.3


11/1/2004 2:59:30 PM    Temperature data are taken from TBSM18 site
Coordinator this week: V.Chohan; MTM support on call :  (16-3602 … Richard).

13 Magnets [ 10 dipoles( 3 with MM)+ 3 SSS +  0 SSSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
HV Insulation re-test on SSS after corrector powering ? (observation Richard )

AOB
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	From 24-10-2004 to 30-10-2004 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBAR3049 
	False 
	24-10-2004 
	25 h.
	28 h.
	26 h.
	20 h.
	1 h.
	28-10-2004 
	4 days
	1 

	2
	TBF1
	MBAL1158 
	False 
	22-10-2004 
	28 h.
	33 h.
	26 h.
	36 h.
	7 h.
	27-10-2004 
	5 days
	2 

	3
	TBE2
	MBAL3232 
	False 
	18-10-2004 
	30 h.
	63 h.
	25 h.
	19 h.
	8 h.
	24-10-2004 
	6 days
	2 

	4
	TBE2
	MBBL1128 
	False 
	24-10-2004 
	23 h.
	41 h.
	14 h.
	14 h.
	8 h.
	28-10-2004 
	4 days
	2 

	5
	TBE1
	MBBL2098 
	False 
	21-10-2004 
	29 h.
	35 h.
	47 h.
	17 h.
	10 h.
	27-10-2004 
	6 days
	9 

	6
	TBD2
	SSS066 
	False 
	14-10-2004 
	78 h.
	28 h.
	76 h.
	11 h.
	36 h.
	24-10-2004 
	10 days
	2 

	7
	TBC2
	MBBL3164 
	True
	17-10-2004 
	33 h.
	87 h.
	66 h.
	13 h.
	5 h.
	26-10-2004 
	9 days
	2 

	8
	TBC2
	MBBL3236 
	True 
	26-10-2004 
	28 h.
	23 h.
	30 h.
	14 h.
	5 h.
	30-10-2004 
	4 days
	2 

	9
	TBC1
	MBBL1150 
	False 
	20-10-2004 
	48 h.
	62 h.
	17 h.
	39 h.
	12 h.
	28-10-2004 
	8 days
	2 

	10
	TBB2
	SSS064 
	False 
	13-10-2004 
	52 h.
	79 h.
	90 h.
	18 h.
	10 h.
	25-10-2004 
	12 days
	2 

	11
	TBB1
	SSS068 
	False 
	21-10-2004 
	40 h.
	18 h.
	71 h.
	5 h.
	7 h.
	27-10-2004 
	6 days
	2 

	12
	TBA2
	MBAL3230 
	True 
	23-10-2004 
	29 h.
	31 h.
	41 h.
	43 h.
	10 h.
	29-10-2004 
	6 days
	2 

	13
	TBA1
	MBAL3226 
	False 
	21-10-2004 
	33 h.
	32 h.
	36 h.
	19 h.
	14 h.
	27-10-2004 
	6 days
	5 


	



[image: image2]
Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3226
	9548 A.
(D1_L)
	10280 A.
(D1_U)
	11729 A.
(D2 - L)
	12136 A.
(D2-L)
	12667 A.
(D1_L)
	12850 A.
[image: image3.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3230
	12599 A.
(D2_U_S01)
	12486 A.
(D1-U)
	[image: image4.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS068
	11914 A.
(QI-12)
	12160 A.
(QI_12)
	[image: image5.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS064
	11703 A.
(QI-12)
	12349 A.
(QE-34)
	[image: image6.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL1150
	12575 A.
(D1_U)
	12850 A.
(No Data)
	[image: image7.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL3164
	10418 A.
(D2_U_S01)
	11824 A.
(D1_S01)
	12850 A.
[image: image8.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBL3236
	9849 A.
(D2-L-S13)
	12208 A.
(D1-L-S01)
	[image: image9.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS066
	12148 A.
(Q_E-12)
	12361 A.
(QE_34)
	[image: image10.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL2098
	9818 A.
(D2-U)
	10967 A.
(D2_L)
	12216 A.
(D2_U)
	12285 A.
(D2-U)
	12260 A.
(D2-U)
	11118 A.
(D2_U)
	11072 A.
(D2_U)
	11197 A.
(D2-U)
	11224 A.
(D2-U)
	[image: image11.png]



	-
	-
	-

	10
	MBAL3232
	10125 A.
(D2_U)
	12022 A.
(D2_U)
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBL1128
	12108 A.
(D2_U)
	12682 A.
(D2-U)
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAL1158
	12342 A.
(D2-L)
	12606 A.
(D2-L)
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR3049
	12084 A.
(D2-L)
	12850 A.
[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


PROBLEMS OF LAST WEEK

	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday _26_M
	TBA2
	Iearth  to high (6-7 mA )recorded during diode test of the MBAL3230.
	MTM
	
	Pierre advised to check the electrical integrity of the magnet (HV test) 

	Tuesday _26_A
	TBF2
	CDAP report shows results not OK : but no step is indicated as “failed”.
	AB/CO?
	HV
	Step(s) not executed in the sequence?

	Wednesday _27_A
	TBD1
	T-coil recordings not correct.
	MTM
	
	Giorgio informed. Sector 04 BCD broken (not use for MQ magnetic measurements). Giorgio found also a problem in the “nappe” which disappeared.

	Wednesday _27_A
	TBD1
	SPA abort during LHC cycle. No cryogenic problem and not detected by the PLC-SE
	
	
	Pierre’s advise: Training quench and repeat LHC cycle.

	Thursday _28_M
	TBD1
	MM3 (full load-line): no Temperature recordings, shaft position alert at high current.

MM4: Correctors measurements stopped : Maryline’s investigation on the MM measurements.
	MTM


	
	Pierre advised to proceed.



	Thursday _28_A
	TBC2
	MM1: not good value for TT821
	MTM
	
	Giorgio investigated the problem: pin broken in the chassis: Chassis changed. 

Then the “0” of the coder not recognized. 1 cable connecting one integrator to the chassis not working +sliding of one aperture (mechanical problem in the TRU). Giorgio solved all the problems but  time lost ~4 h

	Thursday _28_A
	TBC2
	PC blocked in “stand-by mode”’
	AB/PO
	
	Reset the server and re-start?

	Thursday _28_N
	TBB1
	Vacuum leak?
	ICS
	
	ICS asked to connect the blue cable for the anti-cryostat.

	Thursday _28_N
	TBD1
	CDAP, HV test: Stop of the megger during the measurements (rack MXT.MY95)
	
	
	Cable connection between Megger and the bench not working properly.? Richard informed.

	Thursday _28_N
	TBB1
	Not enough wire for SSW measurements. Nitrogen cylinder empty.

Stage B not found.

Wire broken.
	MTM
	
	

	Friday_29_M
	TBB2
	The sequence Ramp_MQ-760 A and Ramp_MQ_11850 A were missing for B2 bench.
	MTM
	
	Giorgio add these sequences.

	Friday_29_M
	TBD1
	No corrector signal seen with shaft measurements.
	MTM
	MMP
	Origin current in one aperture and measurement in the second aperture. Giorgio modified the run type: measurement on both apertures and then OK.

	Friday_29_A
	TBD1
	Motor controller of SSW damaged.
	MTM
	
	Guy informed: The motor controller needs to be switched "ON" after switching "ON" Kepco. After use, we need to switch "OFF" the motor controller before switching "OFF" Kepco. Motor controller has been replaced

	Friday_29_N
	TBB2
	Sextupole measurements at cold not correct, repeated a second time: still outside the window.

Cryo OK lost 2 times
	MTM

ACR
	
	Giorgio informed: the wrong circuit was powered. Octupole powered instead of sextupole one.

	Friday_29_N
	TBD1
	During strength measurements wire broke.

 X-axis result data not found. 
	MTM
	
	New wire and test re-launched.

	Saturday 30_N
	TBB1
	Powering of the EF circuit (octupole) not possible (SSS67)
	ICS?

AB/PO?
	
	Proceed with the rest of the cod test program .AB/PO called monday. Investigation on the run.

	Sunday_31_N
	TBB1
	Wire broke during rot-wire measurements (sextupole) .
	TS/SU
	
	New wire, XY measurements and rot-wire measurements repeated.

	Monday_1_M
	TBD2
	Survey called. No answer.
	TS/SU
	
	Survey came 2h30 after.









Light green : Connect (up to Cryo pump down)	24 h


Blue :  Wait + Cool down				26 h


Ivory: Tests at Cold					36 h


Red :  Wait + Warm up					12 h


Dark Green: Disconnect					12 h








